Author’s response to reviews

Title: Polyphenols of Camellia sinenesis decrease mortality, hepatic injury and generation of cytokines and reactive oxygen and nitrogen species after hemorrhage/resuscitation in rats

Authors:

Mark Lehnert (marklehnert@web.de)
Henrik Lind (henrik.lind@hotmail.de)
Zhi Zhong (zhong@musc.edu)
Robert Schoonhoven (robert.schoonhoven@mpiresearch.com)
Ingo Marzi (marzi@trauma.uni-frankfurt.de)
John J Lemasters (JJLemasters@musc.edu)

Version: 6 Date: 12 August 2010

Author’s response to reviews:

This is the resubmission with the appropriate ABSTRACT.